Author's response to reviews

Title: QuantiFERON-TB-Gold In-Tube test conversions and reversions among Tuberculosis patients and their household contacts in Addis Ababa: a one year follow-up study

Authors:

Mulugeta Belay (mulg2002@yahoo.com)
Mengistu Legesse (mengistu.legessed@aau.edu.et)
Daniel Dagne (danieldagne72@yahoo.com)
Adane Mihret (adane_mihret@yahoo.com)
Yonas Bekele (yoni4m@gmail.com)
Girmay Medhin (gtmedhin@yahoo.com)
Gunnar Bjune (g.a.bjune@medisin.uio.no)
Fekadu Abebe (fekadu.abebe@medisin.uio.no)

Version: 2
Date: 1 August 2014

Author's response to reviews:

Date: August 1, 2014

Cover letter

Editorial Office, BMC Infectious Diseases
Re: QuantiFERON-TB-Gold In-Tube test conversions and reversions among Tuberculosis patients and their household contacts in Addis Ababa: a one year follow-up study

Dear Editor,

On behalf of the authors of this manuscript, I would like to thank you for your quick assessment and comments. As suggested, email addresses of all authors are listed and line numbers are included. Besides, some editorial changes have been made in the manuscript and the updated version of the manuscript as well as the abstract are uploaded.

Kind regards,

Mulugeta Belay, MD, MPhil
Aklilu Lemma Institute of Pathobiology, Addis Ababa University, P.O.Box 1176, Tel: +251 112763091 Mail: mulg2002@yahoo.com,